Author's response to reviews

Title: Analysis of farmers' cognition about new rural cooperative medical system in Liaoning Province, China: a research article

Authors:

xi long pan (xilongpan@126.com)
ying zhang (zhangying2006100@126.com)
zheng wang (zwang@prgs.edu)
qian qian zhao (qianqian_zhao1984@126.com)
ju huang (lihuihuang-1@163.com)
li xu (xuli_530@126.com)

Version: 2 Date: 5 December 2008

Author's response to reviews:

1 We have asked the professional editing company to make our paper better;

2 Some creative parts are also added to the paper, including answering the questions which are not available from the previous studies.

3 We have changed the format in several pages to make sure that they are ready to open to you.